Author's response to reviews

Title: Morning Glory Optic Disc Anomaly in Association With Down Syndrome

Authors:

Ahmet Altun (aaltun06@gmail.com)
Gulengul Altun (gulengulaltun@gmail.com)
Sevda Aydin Kurna (sevdaydin@yahoo.com)
Osman Okan Olcaysu (dr.okan@hotmail.com)
Suat Fazil Aki (suataki@gmail.com)

Version: 4 Date: 27 January 2014

Author's response to reviews:

Dear Editors,

Morning Glory optic disc anomaly has been reported to be associated with some systemic diseases and syndromes. This case report is unique, because this will be the first reported case of Morning Glory optic disc anomaly associated with Down Syndrome in the literature. We tried to present figures and details about the case as much as we were able to. We hope to get a chance to announce this new association in your journal.

Best Regards,

Ahmet ALTUN, MD